Author's response to reviews

Title: Long-term survival after initial hospital admission for peripheral arterial disease in the lower extremities

Authors:

I. Vaartjes (c.h.vaartjes@umcutrecht.nl)
G.J. de Borst (G.J.deBorst-2@umcutrecht.nl)
J B Reitsma (J.Reitsma@amc.uva.nl)
A de Bruin (abun@cbs.nl)
F L Moll (F.L.Moll@umcutrecht.nl)
D E Grobbee (D.E.Grobbee@umcutrecht.nl)
M L Bots (M.L.Bots@umcutrecht.nl)

Version: 5 Date: 26 August 2009

Author's response to reviews:

Dear Sir/Madam,

Hereby I sent you a revised manuscript. I checked the manuscript and I made some small typographical changes. Furthermore, you noticed that reference 47 was not cited and therefore you cited it alongside reference 46 on page 12. This is not correct as reference 47 is cited in the legend of figure 1 and 2. I therefore removed reference 47 from page 12.

kind regards,

Ilonca Vaartjes